EO-1 Weekly Status Week of December 23–January 5, 2011
Day of Year 357 - 005
Mission Day 3695 - 3708
Earth Observing One (EO-1) - General
Scheduled 288 science Data Collection Events (DCEs) the past two weeks.  
INSTRUMENTS

All instruments operated nominally this week 
· Performed instrument calibration

· Conducted ALI internal calibration on December 27 at 00:00z

· Performed instrument decontamination cycles

· Conducted Hyperion deicing on January 3 from 00:25z to 15:15z

· Conducted ALI outgassing on January 3 from 00:35z to 15:25z

Command and Data Handling (C&DH)

EO-1 has continued to experience problems playing back engineering data from the Solid-State Recorders.  Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.
On Tuesday, January 4, the EO-1 Mission Science Office (MSO) conducted a meeting chaired by Betsy Middleton.  The participants were Betsy Middleton, Petya Campbell, Fred Huemmrich, Steve Ungar, David Landis, Nathan Pollack, Qingyuan Zhang, Yen-Ben Cheng, Larry Corp, Robert Knox, Dan Mandl, Stu Frye, Jacqueline LeMoigne, and Joe Young.  

Topics pertinent to EO-1 Operations that were discussed are as follows:

1. Reviewed the EO-1 10 yr celebration events.
2. The upcoming Headquarters Senior Review event was reviewed with the following items of interest discussed and noted:  
· Betsy Middleton reviewed information on the upcoming 2011 Senior Review sent by Cheryl Yuhas. 
·  The proposal is due early March.

· Panel questions from Headquarters are due on April 18.

· The Senior Review will be conducted during the period March-June 2011.
· Extended mission decision announcements will be made in August 2011.

· Stu Frye is to contact USGS to discuss EO-1 benefits to them.

· National and International needs credits and new science activities are to be identified.

· Our proposal draft is to be completed by mid February.
3. The proposed NEREIDS mission, which has cleared the first hurdle for the upcoming Earth Venture (EV-2) call, was discussed by Jacqueline Lemoigne.  She stated that there is to be a meeting with the DIA sometime between 17-26 January to discuss a possible partnership on NEREIDS.  The multi-Directorate Executive Committee is slated to review the NEREIDS EV-2 proposal on January 26 at 4:00.
4. David Landis is to prepare a description of the EO-1 Gulf oil spill data holdings that Betsy will forward to Jon Ransom in response to a request from Peter Hildebrand regarding NASA’s response to the Department of Justice lawsuit concerning the Deep Water Horizon oil spill.
5. There was a discussion of EO-1 scene acquisition issues.  Nathan Pollack reported on acquisition of night time scenes and lunar calibration images for NOAA.  Steve Ungar emphasized the importance of acquiring Hyperion lunar data for calibration purposes to support the NOAA ROLO model and in particular to compute radiances for specific lunar features such as craters.
6. Betsy Middleton discussed GSFC’s FY11 and FY12 budgets for HyspIRI and read a letter from Steve Volz at Headquarters that described the budget amounts.  As a result of this letter, Dan Mandl, Patrick Coronado, and Steve Chien are to stop work on HyspIRI funded Direct Broadcast development.
7. Betsy stated that this year’s GSFC sponsored HyspIRI Science Workshop will be held the middle week in May.
8. There was a discussion of image data atmospheric correction issues that included the need to obtain a “water version” of ATREM from NRL and also obtain the source code so that it can be complied on the new server in Bldg 33.
9. Betsy reported on a statement by Garik Gutman on the current ROSES 2010 Research Announcement for Land-Cover and Land-Use Change (Step 1 and Step 2) in that it emphasizes inputs by young scientists.  Also, Garik Gutman will not develop a Global Land Survey (GLS) after the current 2010 version is released.  Therefore, no further acquisitions for a GLS will be needed from EO-1.
10. Stu Frye reported that Disaster Monitoring activity is continuing while we are repairing the GeoBPMS tasking software.  The GeoBPMS maintenance task has been authorized to proceed on the new SES contract.
11. Stu Frye and Fritz Policelli recently won a ROSES proposal award on a study of Terra/Aqua/RADARSAT combined optical/radar data to produce daily flood maps in the Caribbean.

12. Betsy proposed that at the IGARSS meeting in Japan in August 2011 there be an EO-1 Special Session.
13. David Landis showed a series of what is considered the most important EO-1 images acquired the last two years.

14. Betsy stated that there is a call to review a ROSES new U.S. Carbon Cycle Science Plan with inputs due February 15.
15. The next MSO meeting will be on January 25.
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Continued nominal spacecraft state of health

· Testing additional Norway (Svalbard) ground systems to replace SG-1 until it is green

· SG1/SG22 Hybrid is now certified for S-band supports

· Testing SG1/SG22 hybrid for S/X Band

· This station is using the SG1 back end computer system hooked up to the SG22 antenna

· SGS testing will resume once EO-1 has been kept in SGS schedule 

Mission Planning

No change from last report.
Flight Dynamics 

· Problem with PF1 tracking data discovered

· The problem is with the files sent by PF1 contains the wrong DOY, and the sent files are 0KB

· Currently working with PF1 personnel to resolve the issue

Trending
No change from previous week.
System Administration
· Began testing of the ClamAV Anti-Virus software that is to be installed on all EO-1 Linux computers.

· Installed Linux on a spare computer to act as an initial test bed.

· Installed ClamAV on the computer.

· Wrote down notes from the installation, configuration, and research for future installs.

· Ran a few test scans of the system. 

· These tests seem to have been successful.

· Will continue to run manual tests as well as begin to research automated scanning.

· Still researching the process for updating virus definitions.

· Received a request from one of the EO-1 scientists for three additional computers to be able to retrieve science data from our data processing computer.

· Requested an Open IONet Firewall rule update (OGRS #06-0073) to add port 22 access from the three new computers.

· This rule was approved but has not been implemented yet.

· Once the Open IONet Firewall rule is implemented I will update the EO-1 mission firewall.

· Continued creating images of operational computers.

· Imaging the ITPS trending computer.

· Have been in contact with DTN developers regarding their upcoming tests.

· Exchanged necessary information for the connectivity test between the Bldg 25 DTN node and the DTN node in our MOC.

· Creating a MOCR for approval of the Bldg 25 connection to the MOC.

GROUND AND SPACE NETWORK

Station Downtimes

· SG-1 has been declared red until further notice.
· MGS red for planned maintenance.
Operational Discrepancies
· On WGS pass on December 27 at 01:26z the 11 Meter ACU crashed during support resulting in partial loss of ALI internal calibration data
· SGS pass on December 27 at 18:01z received garbled data possible because of S-Band only tracking due to heavy snow resulting in loss of 2 images
· SGS passes below were lost due to EO-1’s removal from SGS schedule:

· December 27 at 21:17z with loss of 4 images
· December 28 at 10:31z with loss of 4 images
· December 28 at 13:47z with loss of 3 images
· December 28 at 17:01z with loss of 4 images 

· December 29 at 12:47z with loss of 1 image
· December 29 at 17:39z with loss of 1 image 

· December 30 at 16:39z with loss of 2 images
· January2, 2011 at 10:25z with loss of 4 images
UPCOMING EVENTS

· Next inclination maneuvers to be in beginning to mid February and will be approximately 600 sec in duration
· DTN testing will resume sometime the last week in January

Imagery Status

Scenes and Engineering Cals planned for two weeks of December 23, 2010-January 5, 2011     288
Total scenes and engineering calibrations planned for entire mission – approximately 57,891
Total Scenes:  ALI scenes in the Level 0 archive              53,207                (as of January 5, 2011)

                         Hyperion scenes in the Level 0 archive    52,954               
Publications and Presentations Status 

430 publications 

295 presentations

53 articles and press releases
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